UC San Diego UC San Diego Previously Published Works

Title

Biotic and Human Vulnerability to Projected Changes in Ocean Biogeochemistry over the 21st Century

Permalink https://escholarship.org/uc/item/2w29w8cn

Journal PLOS Biology, 11(10)

ISSN 1544-9173

Authors

Mora, Camilo Wei, Chih-Lin Rollo, Audrey <u>et al.</u>

Publication Date 2013-10-01

DOI

10.1371/journal.pbio.1001682

Peer reviewed

Biotic and Human Vulnerability to Projected Changes in Ocean Biogeochemistry over the 21st Century

Camilo Mora¹*, Chih-Lin Wei², Audrey Rollo³, Teresa Amaro⁴, Amy R. Baco⁵, David Billett⁶, Laurent Bopp⁷, Qi Chen¹, Mark Collier⁸, Roberto Danovaro⁹, Andrew J. Gooday⁶, Benjamin M. Grupe¹⁰, Paul R. Halloran^{11,12}, Jeroen Ingels^{13,14}, Daniel O. B. Jones⁶, Lisa A. Levin¹⁰, Hideyuki Nakano¹⁵, Karl Norling¹⁶, Eva Ramirez-Llodra¹⁷, Michael Rex¹⁸, Henry A. Ruhl⁶, Craig R. Smith¹⁹, Andrew K. Sweetman²⁰, Andrew R. Thurber²¹, Jerry F. Tjiputra²², Paolo Usseglio^{23,24}, Les Watling²³, Tongwen Wu²⁵, Moriaki Yasuhara²⁶

1 Department of Geography, University of Hawaii, Honolulu, Hawaii, United States of America, 2 Ocean Science Centre, Memorial University of Newfoundland, St. John's, Newfoundland, Canada, 3 Pacific Islands Fisheries Science Center, Honolulu, Hawaii, United States of America, 4 Norwegian Institute for Water Research, Bergen, Norway, 5 Florida State University, Tallahassee, Florida, United States of America, 6 National Oceanography Centre, University of Southampton Waterfront Campus, Southampton, United Kingdom, 7 Institut Pierre Simon Laplace/Laboratoire des Sciences du Climat et de l'Environnement, Centre National de la Recherche Scientifique, Gif sur Yvette, France, 8 The Centre for Australian Weather and Climate Research, Commonwealth Scientific and Industrial Research Organisation Marine and Atmospheric Research, Aspendale, Victoria, Australia, 9 Department of Life and Environmental Sciences, Polytechnic University of Marche, Ancona, Italy, 10 Center for Marine Biodiversity and Conservation, Scripps Institution of Oceanography, La Jolla, California, United States of America, 11 Met Office Hadley Centre, Exeter, United Kingdom, 12 College of Life and Environmental Sciences, University of Exeter, Exeter, United Kingdom, 13 Marine Biology Research Group, Biology Department, Ghent University, Ghent, Belgium, 14 Plymouth Marine Laboratory, Plymouth, United Kingdom, 15 Meteorological Research Institute, Tsukuba, Japan, 16 Norwegian Institute for Water Research, Oslo, Norway, 17 Institut de Ciències Marines, Consejo Superior de Investigaciones Científicas, Barcelona, Spain, 18 Department of Biology, University of Massachusetts, Boston, Massachusetts, United States of America, 20 International Research Institute of Stavanger, Thormøhlensgate, Bergen, Norway, 21 College of Earth, Ocean, and Atmospheric Sciences, Oregon State University, Corvallis, Oregon, United States of America, 22 Uni Climate, Uni Research, Bergen, Norway, 23 Department of Biology, University of Hawaii at Manoa, Hawaii, United States of America, 2

Abstract

Ongoing greenhouse gas emissions can modify climate processes and induce shifts in ocean temperature, pH, oxygen concentration, and productivity, which in turn could alter biological and social systems. Here, we provide a synoptic global assessment of the simultaneous changes in future ocean biogeochemical variables over marine biota and their broader implications for people. We analyzed modern Earth System Models forced by greenhouse gas concentration pathways until 2100 and showed that the entire world's ocean surface will be simultaneously impacted by varying intensities of ocean warming, acidification, oxygen depletion, or shortfalls in productivity. In contrast, only a small fraction of the world's ocean surface, mostly in polar regions, will experience increased oxygenation and productivity, while almost nowhere will there be ocean cooling or pH elevation. We compiled the global distribution of 32 marine habitats and biodiversity hotspots and found that they would all experience simultaneous exposure to changes in multiple biogeochemical variables. This superposition highlights the high risk for synergistic ecosystem responses, the suite of physiological adaptations needed to cope with future climate change, and the potential for reorganization of global biodiversity patterns. If co-occurring biogeochemical changes influence the delivery of ocean goods and services, then they could also have a considerable effect on human welfare. Approximately 470 to 870 million of the poorest people in the world rely heavily on the ocean for food, jobs, and revenues and live in countries that will be most affected by simultaneous changes in ocean biogeochemistry. These results highlight the high risk of degradation of marine ecosystems and associated human hardship expected in a future following current trends in anthropogenic greenhouse gas emissions.

Citation: Mora C, Wei C-L, Rollo A, Amaro T, Baco AR, et al. (2013) Biotic and Human Vulnerability to Projected Changes in Ocean Biogeochemistry over the 21st Century. PLoS Biol 11(10): e1001682. doi:10.1371/journal.pbio.1001682

Academic Editor: Georgina M. Mace, University College London, United Kingdom

Received March 26, 2013; Accepted September 3, 2013; Published October 15, 2013

Copyright: © 2013 Mora et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This study was funded by the University of Hawaii Sea Grant (http://seagrant.soest.hawaii.edu/). We thank the Norwegian Research Council (www. forskningsradet.no/en/) for funding a workshop on Climate Change Stress on Deep-Sea Benthic Ecosystems (CLIDEEP), the Sloan Foundation through the Census of Marine Life Program (www.coml.org) and the Total Foundation (http://foundation.total.com/) for funding several meetings as part of the International Network for Scientific Investigations of Deep-Sea Ecosystems (INDEEP, www.indeep-project.org) and the First Global Synthesis of Biodiversity, Biogeography and Ecosystem Function in the Deep Sea (SYNDEEP) that led to the idea of this paper. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

Abbreviations: CMIP5, the Coupled Model Intercomparison Project Phase 5; GDP, gross domestic product; POC flux, particulate organic carbon flux; RCP, representative concentration pathway.

* E-mail: cmora@hawaii.edu

PLOS BIOLOGY

Author Summary

Climate change caused by human activity could damage biological and social systems. Here we gathered climate, biological, and socioeconomic data to describe some of the events by which ocean biogeochemical changes triggered by ongoing greenhouse gas emissions could cascade through marine habitats and organisms, eventually influencing humans. Our results suggest that the entire world's ocean surface will be simultaneously impacted by varying intensities of ocean warming, acidification, oxygen depletion, or shortfalls in productivity. Only a very small fraction of the oceans, mostly in polar regions, will face the opposing effects of increases in oxygen or productivity, and almost nowhere will there be cooling or pH increase. The biological responses to such biogeochemical changes could be considerable since marine habitats and hotspots for several marine taxa will be simultaneously exposed to biogeochemical changes known to be deleterious. The social ramifications are also likely to be massive and challenging as some 470 to 870 million people - who can least afford dramatic changes to their livelihoods - live in areas where ocean goods and services could be compromised by substantial changes in ocean biogeochemistry. These results underline the need for urgent mitigation of greenhouse gas emissions if degradation of marine ecosystems and associated human hardship are to be prevented.

Introduction

As CO₂ and other greenhouse gas emissions continue to rise, ocean biogeochemistry is being altered in ways that could potentially impact nature and mankind. Atmospheric CO₂ concentrations have already risen to $\sim 400 \text{ ppm}$ from \sim 280 ppm in pre-industrial times and could rise to between 550 and 900 ppm by 2100, depending upon the emission scenario [1-7]. In the marine realm, the surplus of CO₂ has been associated with ocean warming from the greenhouse effect [1] and acidification caused by the fact that approximately 25% of the annually emitted CO_2 enters the ocean, where it reacts with water to produce carbonic acid, thereby reducing pH [6-8]. Ocean warming and other climatic changes can trigger additional responses in connection to ocean circulation and stratification, which in turn reduce oxygen concentration [9,10] and primary productivity [11] (additional responses may include sea-level rise and extreme weather events, which we do not analyze here but that certainly will add to the stress likely to be exerted by greenhouse gas emissions [10]). Several analyses predict that, by the year 2100, depending on the emission scenario, surface ocean temperature could increase by 2 to 3° C [9], pH decline by over 0.2 units [6,7], oxygen concentration decrease by 2% to 4% [9], and ocean productivity by 2% to 20% [11], from current values. The magnitude of these changes would be unprecedented in the Earth's history during the last 20 million years [12,13].

Species are adapted to their environment, and therefore shifts in environmental parameters can induce considerable change in species fitness and trigger additional responses in community composition, functioning, and overall biodiversity [2,3,9–11,14– 16]. Ocean warming, acidification, oxygen depletion, and reduction in primary production have all been highlighted as potentially having negative biological consequences [2,3,9–11,14– 16]. Changes in temperature, for instance, can affect metabolism, reproduction, and survival [10,17], which is already evident in multiple shallow and deep-sea ecosystems [2,18]. Parameters related to food supply, such as primary productivity and sinking organic-carbon flux, and dissolved oxygen can influence metabolism, body size, reproduction, and thus control, in part, the biomass that can be sustained in any given area of the ocean [19]. Moreover, depending on the magnitude of shifts in biogeochemical parameters and/or their proximity to physiological thresholds, these changes can make entire areas essentially unsuitable for metazoans (except for some meiofaunal organisms, as well as viruses, prokaryotes, and certain protists [20–23]). There is already evidence that oxygen minimum zones have increased in vertical extent over recent decades, with important consequences for ecosystems and coastal communities [24]. Likewise, pH can influence rates of calcification and several other physiological processes [10,15,25,26]. Co-occurring changes in biogeochemical parameters could also accelerate biological responses, either additively or synergistically [10,27-30]. Warming, for instance, can increase metabolism but, if combined with a reduction in dissolved oxygen and food availability, it could also lead to considerable reductions in body size [31], survival, and synergistic responses of ecosystems [32] and cause range expansions or contractions [2,10,17]. Studies on marine invertebrates have also revealed that embryos that survive exposure to warming may later die as larvae if exposed to acidification [33]. This is not to say that all species will be impacted negatively. Some species may expand to new areas or thrive in areas where they were once rare. It is certain, however, that biogeochemical changes in the ocean, especially their co-occurrence, have considerable potential to reorganize patterns in biodiversity, body size, and abundance (Table 1). Additionally, the number of species within ecosystems, variations in life histories, and susceptibility to climate change among species suggest that ecosystem responses to ocean biogeochemistry change are likely to be varied and highly idiosyncratic (Table 1).

Socioeconomic systems can also be sensitive to ocean biogeochemical changes, depending upon the exposure of ocean goods and services to environmental change, human dependence on affected services, and social adaptive capacity [34-38]. Examples of the goods and services likely to be impacted by ocean climate change are diverse. Ocean warming and acidification, for instance, are causing a new set of conditions that are very close to the tolerance thresholds of corals, making them vulnerable to massive bleaching and mortality when long-term trends related to climate change are "added" to natural variability. The decay of coral reefs could potentially impair their ability to deliver goods and services such as fisheries, tourism, coastal protection, and in some cases aesthetic and spiritual values [35,37], which have been grossly valued at over US\$375 billion annually [39]. Likewise, future changes in ocean temperature are expected to cause a redistribution in the global diversity of cetaceans [40], which in turn could impact local economies that rely on tourism or the fishing of these species. A similar example is the effect of ocean climate change on the world's fisheries, where a combination of warming, oxygen depletion, and reduction in primary productivity can induce changes in body size [31], abundance, and distribution of exploited species [41,42]. These would add to the ongoing decline of fisheries yields, which are considerable sources of food, revenues, and jobs [36,42,43]. Shifts in the distribution and abundance of species could also bring new opportunities for local communities, although adaptability (e.g., flexibility and responsiveness) will be needed to realize any potential benefits [38]. However, the vulnerability of societies to the changes in ocean goods and services ultimately depends on the balance among exposure to environmental change, human dependency on impacted goods and services, and social adaptability

Table 1. Likely biological responses to changes in ocean biogeochemistry.

	Temperature	рН	Oxygen	Productivity
Body size and growth	Due to temperature control over metabolism [60], everything else being equal, warming should reduce growth and body size [31,61,62]. In some regions, warming of extreme cold places could enhance individual body growth [63].	Acidification may reduce skeletogenesis [33,64] and increase metabolic costs of calcification [32], although some taxa are resistant [65] and some plants may benefit [66] (but see [67]). CO ₂ can increase in the blood (i.e., hypercapnia) reducing growth [33,68–70].	Hypoxia (reduced oxygen) should reduce growth and body size [71–73]. Oxygen concentration also exerts a strong control over calcification rates of corals [74].	Growth and body size should decline with lowered productivity [19,31,75–79]. Changes in life- history strategies of abyssal macrofauna may be related to changes in surface productivity [80].
Survival and abundance	In some taxa, thermal tolerance thresholds could be surpassed by warming leading to excessive mortality (3,81–83), especially if in interacting with other stressors [29,84]. Warming thus reduces abundance [83,85–87] and may enhance diseases [88–93].	Acidification increases mortality in selected adult [94] and juvenile [95–98] marine invertebrates [33] and plants [67]. Abundance can decline among producer species [67] (but see [66,99]).	Hypoxia causes mortality in most large eukaryote species [23,71,84,100], and anoxia (complete lack of oxygen) could cause extinction in macro- and megafauna [71,101–104]. Hypoxia may enhance dominance by some taxa that are hypoxia tolerant [103,105,106] or that are released from ecological interactions [16,71,107,108].	Mortality of benthic invertebrates is generally higher with reductions in food supply [83]. Reduced productivity could reduce abundance [75,83,108–114] and lead to dominance shifts from large to small taxa [115].
Range and distribution	Warming could cause range shifts poleward and to deeper waters [116–119], which in turn could affect the strength of ecological interactions [120], gene flow, and rates of evolution [121]. Warming also reduces habitat suitability for species that do not shift ranges [122].	Reduced calcium carbonate saturation could prevent calcification and growth and thus lead to the disappearance of calcifying species from certain shallow [3,123] and deep-sea [124] areas.	Some taxa may disappear from hypoxic waters [24,71,103,125–129] but others may appear and thrive [24,125,128]. Some evidence exists for increased endemism among benthic foraminifera in core regions of oxygen minimum zones [130].	Certain species are unlikely to maintain their distribution in food- limited areas of the seafloor [131].
Species richness	Theory suggests a positive relation between richness and temperature [132–135], which is confirmed in several marine studies [54,117,136,137]; although some regions and/or taxa fail to show a relationship [138].	Acidification will likely lead to loss of species [94,139,140].	Diversity declines as oxygen declines for protists [16,23,101], meiofauna [16], macrofauna, and megafauna [23,24,71,101,125].	Richness shows a unimodal [83,112,114,131] or no [137,138] relationship with proxies of food supply. Productivity seasonality may negatively affect diversity [141,142]. Eutrophication causes diversity decline via hypoxia and anoxia [16].
Functioning	Ecosystem malfunctioning could be extensive if key-stone species are affected [3,55,56,120,122]. Trophic cascades (e.g., rise of jellyfish) could also occur [105].	Acidification can affect nutrient cycling [140,143], while reduced calcification can reduce sinking rates and carbon export fluxes to the seafloor via less mineral ballast [144].	Carbon cycling could shift from metazoans to benthic foraminifera [145] and microbiota [20,145] in suboxic and anoxic zones. Hypoxia can reduce colonization, recovery, and resilience [146].	Reduced food supply can reduce carbon cycling [19,147,148], modify food-web structures [114], and cause shifts from macrofaunal- to microbial-dominated nutrient cycling [75,149,150].

doi:10.1371/journal.pbio.1001682.t001

[34,35,44,45]. In that context, we are aware of two relevant studies analyzing social vulnerability to ocean climate change over large spatial scales: one for fisheries [34] and the other for coral reefs [35]. Given the limited availability of ocean climate projections at the time, the former study used projected mean surface air temperature to 2050 as the underlying indicator of exposure to climate change, while the latter study focused on five countries of the Western Indian Ocean and used thermal stress on coral reefs as a proxy of climate change. As far as we are aware, more detailed studies connecting the exposure to several and co-occurring stressors of climate change with a variety of ocean goods and services at the global scale are lacking.

As indicated above, we have a relatively good understanding of the potential changes in ocean biogeochemical parameters expected under different greenhouse gas scenarios [7,9,11,46], and conceptually we know some of the mechanisms through which ecological and social systems may be impacted by such changes. However, we lack a synthetic global quantification of the simultaneous projection of biogeochemical changes on the ocean and how they may pertain to marine biota and people worldwide. To address this gap, we compiled all available data generated by Earth Systems Models as part of the Coupled Model Intercomparison Project Phase 5 (CMIP5) to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [47] to assess the extent of co-occurrence of changes in temperature, pH, oxygen, and primary productivity. We complemented the analysis by assembling global distribution maps of 32 marine habitats and biodiversity hotspots to assess the potential vulnerability of biological systems to co-occurring biogeochemical changes in the ocean. Finally, we used available data on human dependency on ocean goods and services and social adaptability to quantify the vulnerability of coastal people to projected ocean biogeochemical change. We would like to emphasize that our results primarily concern the vulnerability of biological and social systems resulting from their exposure to projected anthropogenic ocean biogeochemical change, while cautioning that, although biotic and social responses will certainly occur, the type and magnitude of such responses will be difficult to predict.

Results and Discussion

Important contributions have been made to understand future projections in the ocean biogeochemical parameters analyzed here [9,11,15,46]. We repeated the collection of such projections for the purposes of identifying patterns of co-occurrence in biogeochemical variables and to quantify sources of error due to model accuracy and precision. Earth Systems Models in the CMIP5 improve upon earlier models and runs by incorporating better knowledge of the climate, improved computational capability, and CO₂ pathways that use more detailed and up-to-date data and integrate multiple forcing agents of climate change [48]. Additionally, as noted below, multimodel averages were always more accurate than individual models, further justifying the assembly of biogeochemical projections based on all available Earth System Models. Although these data on ocean biogeochemical parameters represent an important component of our study, our main goals are to identify how their patterns of co-occurrence may pertain to marine biota and thereby social systems worldwide that rely on marine biodiversity goods and services.

Earth System Models Precision and Accuracy

The reliability of climate change projections is primarily determined by the skilfulness with which climate models are able to predict the climate [49,50]. Climate model realism has improved over recent years owing to increased computing power, better scientific understanding of Earth System processes, and the ability to integrate atmosphere, ocean, land, and sea-ice components of the climate system [49-51]. However, our theoretical understanding of the climate system is still incomplete and a myriad of unresolved differences exist among models (e.g., spatial and temporal resolution, numerical solution techniques, process parameterizations, and complexity of atmospheric convection, carbon cycle coupling, ocean mixing, unresolved attributes of the biosphere, etc. [49]). As a result, one of the major motivations in climate research has been to quantify the agreement among models as well as between models and actual climate observations. To address these standing concerns, we measured two proxies for model precision and accuracy. Accuracy was defined as the proximity of the model projections to actual data and precision as the standard deviation among the projections of all models. Of course, the availability of actual observations is restricted to recent times and so we assume that a model that accurately simulates present climate will produce better projections of future climates [51]

We found that the average of all models was always closer to actual observations than any model was individually (Tables S1, S2). Thus, errors in precision were often larger than those in accuracy (Figure 1). That is, there were often large differences among the suite of model predictions, but their multimodel average was often closer to actual observations (Tables S1, S2, Figure 1). We also found that the accuracy of the multimodel average varied by parameter and ocean domain. Specifically, there was a stronger predictability of temperature, oxygen, and pH at the ocean surface and a lower predictability of phytoplankton carbon concentration and of all parameters at the seafloor (Figure 1; complete results and details of accuracy and precision plus Taylor diagrams are presented in Tables S1, S2). This low predictability may emerge from the limited availability of actual observations [this may be the case for "phytoplankton carbon concentration" and "particulate organic carbon flux," which are modeled from other parameters, and there is often a significant disagreement among available products of such parameters (Page 5 in Table S2)] [11] and the inevitable complexity of deep-water processes, which may remain poorly modeled by Earth System Models. With these considerations in mind, we used results based on the upper layer of the ocean and the multimodel average, unless otherwise indicated.

It is worth noticing that discrepancies between Earth System Models outputs and present-day climate observation are partly due to the fact that these models simulate their own internal climate variability (i.e., complex, nonlinear interactions among different components such as atmosphere, ocean, ice, physics, biogeochemistry, etc.) rather than those observed in reality. Thus, a perfect match between any individual model output and observations is unlikely for all places and times. However, these offsetting errors between a given global model and current-day observations have been found to be ameliorated by averaging the output of multiple models (e.g., this study, [52]). This property of multimodel averaging is likely to be just as useful in future climate projections, which highlight the key reason for using the broad range of available models in future predictions of the climate, including those models with moderate capacity to predict current observations [50,52].

Future Projections in Ocean Biogeochemistry

In this study, we analyzed ocean biogeochemical projections under two alternative pathways in which CO_2 concentrations could increase to 550 and 900 ppm by 2100 (as reference, atmospheric CO_2 concentrations are now at ~400 ppm from 280 ppm in pre-industrial times; see Figure S1 [1,48,53]). These two scenarios are based on Representative Concentration Pathways 4.5 (RCP45) and 8.5 (RCP85) and represent alternative mitigation efforts between a concerted rapid CO_2 mitigation and a "business-as-usual" scenario, respectively [48]; there is a more aggressive mitigation scenario called RCP26, which we do not use because it was not consistently used among models and some consider it realistically unattainable (see Figure S1).

Projections of biogeochemical parameters under RCP45 and RCP85 were variable in magnitude among analyzed Earth System Models (semitransparent lines in Figure 2E-H) but followed remarkably similar trends overall (solid lines in Figure 2E-H, Table S1). By 2100, global averages for the upper layer of the ocean could experience a temperature increase of 1.2 to 2.6°C (Figure 2E, Table S3), a dissolved oxygen concentration reduction of 0.11 to 0.24 ml l⁻¹ (i.e., a $\sim 2\%$ to 4% reduction of current values, Figure 2F, Table S3), a pH decline of 0.15 to 0.31 (Figure 2G, Table S3), and a diminished phytoplankton concentration of 0.001 to 0.003 mg C l^{-1} (i.e., a ~4% to 10% reduction of current values, Figure 1H, Table S3) according to RCP45 and RCP85, respectively. In contrast, the world's seafloor was projected to experience smaller changes in temperature and pH (i.e., warming of 0.20 to 0.31°C and acidification of 0.03 to 0.04 pH units) but larger reductions in particulate carbon flux (i.e., food supply) reaching the seafloor (i.e., particulate carbon flux will decline 0.18 to 0.36 mg C m⁻² y⁻¹ or 6% to 13% reduction of current values, Table S3); reductions in dissolved oxygen will be similar to those observed at the sea surface (i.e., oxygen will decline by 0.11 to 0.14 ml l^{-1} compared to current values, Table S3); all values are according to RCP45 and RCP85, respectively.

By 2100, projected changes in temperature, dissolved oxygen, pH, and primary food supply vary significantly among regions (Figure 2A–D). For the ocean surface, the smallest projected changes for pH are in the tropics, for temperature and productivity in temperate regions, and for oxygen in the Southern



Figure 1. Accuracy and precision on future ocean biogeochemical projections. Plots A–D refer to sea-surface parameters; plots E–H to seafloor parameters. These plots illustrate the number of 1 km² cells by their projected change to the year 2100 under the RCP45 (blue lines), RCP85 (red lines), errors in accuracy (green lines), and precision (purple lines). Accuracy was defined as the difference between multimodel average projections and actual data and precision as the standard deviation among the projected changes in surface temperature, oxygen, and pH. Note that in those cases, accuracy (green lines) is centered to zero, meaning that for the great majority of cells the multimodel average prediction was identical to actual observations. Errors in precision were often larger, but they are added to both sides of the projections, meaning that they will broaden expected



Figure 2. Future biogeochemistry change in the world's oceans. Plots A–D show the spatial difference between future (i.e., the average from 2091 to 2100) and contemporary (i.e., the average from years 1996 to 2005) values under the RCP85 scenario (decadal averages were chosen to minimize aliasing by interannual variability; beside each color scale we provide the absolute change, whereas the numbers on top indicate the rescaled values; complete results for the RCP85 and RCP45 for the ocean surface and floor are shown in Figure S2). Plots E–H show the global average change relative to contemporary values under the RCP45 and RCP85 at the ocean surface and seafloor; semitransparent lines are the projections for individual models.

doi:10.1371/journal.pbio.1001682.g002

Ocean (Antarctica). At the seafloor, all variables analyzed experienced the largest changes along continental margins, with decreasing oxygen being common over larger areas of the world's seafloor, particularly at the poles (Figure S2). In general, however, with the exception of the Antarctic and small areas in the South Pacific and North Atlantic, most of the world's oceans will be simultaneously exposed to change in all parameters (Figures 3–4, Figure S2). With the exception of productivity and all parameters at the seafloor, current errors in accuracy and precision of the Earth System Models are of insufficient magnitude to offset projected changes; that is, projected changes in temperature, oxygen, and pH in the upper ocean layer were larger than their errors in accuracy and precision, meaning that trends in these three parameters are robust and are unlikely to be reversed by current sources of model errors (Figure 1, Table S2).

To identify patterns of co-occurrence in biogeochemical changes, we differentiate changes in biogeochemistry that are negative (i.e., warming, acidification, oxygen depletion, and primary food reduction) from those that are positive (i.e., cooling, basification, oxygenation, and productivity increase). Note that the terms "negative" and "positive" are used to indicate the direction of biogeochemical changes, not their potential effects upon



Figure 3. Co-occurring ocean biogeochemical changes to the year 2100 under the RCP85. For these plots, we separated absolute changes shown in Figure 2A–D between those that will be positive (i.e., cooling, basification, oxygenation, and productivity increase; Plots A–E) and negative (i.e., warming, acidification, oxygen depletion, and primary food reduction; Plots F–J). Resulting absolute changes were scaled between 0 and 1 (Plots B–E, G–J), 0 being zero absolute change and 1 being the extreme 97.5% observed value globally. The resulting scaled scores from each variable were added to provide a global composite map of co-occurring positive (Plot A) and negative (Plot F) changes in ocean biogeochemistry. These cumulative change maps ranged from 4 (i.e., the maximum predicted change in all four parameters occurred in that cell) to 0 (i.e., no negative or positive change in any of the four parameters occurred in that cell). The results for the RCP45 at the ocean surface and both RCPs for the seafloor are presented in the Supporting Information section.

doi:10.1371/journal.pbio.1001682.g003



Figure 4. Exposure of the world's oceans to co-occurring changes in ocean biogeochemistry to the year 2100. (A–B) are the cumulative percentage of cells globally exposed to the composite score of co-occurring ocean biogeochemistry changes (see Figure 3 for details). (A) is for negative and (B) for the positive biogeochemistry changes. (C–D) is the discrimination of total ocean cells globally exposed to negative (C) and positive (D) changes in each variable and the composite score. doi:10.1371/journal.pbio.1001682.g004

biodiversity or social systems. The resulting values were then scaled from 0 to 1 (i.e., 0 meaning no change and 1 the upper 97.5% most extreme absolute change predicted in the world). The scaled-scores for each biogeochemical parameter were added to generate a composite global map of "negative" and "positive" changes in ocean biogeochemistry (Figure 3). The composite global scores were differentiated between "positive" and "negative" changes to avoid neutralization of biogeochemical changes (e.g., a cell with a warming score of -1 and a productivity increase score of 1 will yield a composite global score of 0, which would be confounded with no change). Additionally, separation of the global composite scores into positive and negative changes allows a better appreciation of the preponderance of the directions of biogeochemical change in the world's oceans.

The results of this analysis indicate that the entire ocean surface will be impacted by warming, acidification, or reductions in oxygen and productivity (Figure 4A,C)—over 99% by the largest negative change in at least one full parameter (Figure 4A). In contrast, only oxygen and productivity will experience positive changes at the surface over a small fraction (Figure 4D) of the polar regions (Figure 3C,E); almost no place in the world's ocean surface will face cooling or pH increase (Figure 3B,D). Cooccurring negative changes will also occur extensively over the world's ocean seafloor (Figure 4C,D), although the magnitude of such changes will be smaller: only about 20%–27% of the ocean's seafloor will be exposed to the largest negative change projected in more than one biogeochemical parameter (Figure 4A). Patterns of co-occurrence in biogeochemical parameters were very similar between the RCP45 and RCP85 (Figure 4).

Biological Exposure to Ocean Biogeochemistry Change

By overlaying the global distribution of marine habitats and hotspots of biodiversity for individual taxa with the projected changes in temperature, oxygen, pH, and primary food supply, we found that, to varying degrees, all projected biogeochemical changes will occur simultaneously within all habitats and biodiversity hotspots (Figure 5; Table S4 provides detailed statistics for the change in each parameter at each marine habitat and biodiversity hotspot and sources of error owing to accuracy and precision in the Earth System Models). Among marine habitats, the smallest absolute changes in biogeochemical parameters are expected to occur in deep-sea habitats (e.g., soft- and hard-bottom benthos, seamounts, and vents; Figure 5A, Table S4), whereas the largest changes will likely occur in shallow-water habitats like coral and rocky reefs, seagrass beds, and shallow soft-bottom benthos (Figure 5A, Table S4). Like the biota, biodiversity hotspots (i.e., areas with high numbers of species of a particular taxon [54]) will also be differentially stressed by ocean biogeochemistry change



Figure 5. Future ocean biogeochemistry change on marine habitats and biodiversity hotspots. Here we show the mean (horizontal dashes) and standard deviation (curved lines) of the absolute change in each parameter projected to the year 2100 for each marine habitat (Plot A) and biodiversity hotspot for individual taxa (Plot B). A hotspot is defined as the top 10% most diverse (in number of species) areas on Earth where the given taxa are found [54]. In both plots, values for each parameter are color-coded according to the left-hand axes. Values to the left and right of each habitat or hotspot indicate the expected results according to RCP85 and RCP45, respectively. Data on marine habitats were obtained mainly from

Halpern et al. [59]; additional sources are indicated in the Table S6; data on biodiversity hotspots were obtained from Tittensor et al. [54]. Complete results of the exposure of each habitat and hotspots to all parameters as well as the sources of error due to accuracy and precision are presented in

Table S4. Particulate organic carbon flux (or simply carbon flux in the legend) applies only to seafloor habitats.

(Figure 5B). Among biodiversity hotspots analyzed in this study, the smallest cumulative exposure to future biogeochemical change is projected to occur in hotspots of mangrove and coral reef species, whereas the largest exposure will occur in hotspots of euphausiid (i.e., krill; a crucial component of food webs at mid and

high latitudes), cetacean, squid, and pinniped species (Figure 5B, Table S4).

For the purpose of assessing the co-occurrence of biogeochemical change, we considered all absolute changes in an additive manner; however, this is not to say that biological responses will

doi:10.1371/journal.pbio.1001682.g005



Figure 6. Vulnerability of humans to projected ocean biogeochemistry change. This plot illustrates the total number of people likely to be vulnerable through exposure to ocean biogeochemistry change according to RCP45 (Plot A) and RCP85 (Plot B). Numbers in the plot are in billions (summations may not be exact owing to rounding). Categorization of people according to their levels of exposure to biogeochemical changes, dependency on ocean goods and services, and social adaptability is described in the main text. doi:10.1371/journal.pbio.1001682.g006

follow an additive or linear response to such changes. Realistically, empirical data are unavailable for a sufficient number of species to predict the biological responses of an entire ecosystem to the exposure of biogeochemical change in the ocean (i.e., given variations in physiological adaptations, tolerance thresholds, nonlinear responses, ecological interactions, and resulting cascade effects, etc.). Even broad generalizations could be prone to limitations. For instance, it is often argued that diverse ecosystems can be resilient to climate change as redundancy in species functions could allow the buffering of species lost by climate change. This, in itself, implies a change in community structure [55], although empirical evaluation of this idea has suggested that, perhaps due to strong niche specialization, diverse ecosystems may actually exhibit reduced functional redundancy and be particularly prone to disturbances [56].

Despite our inability to predict the type and magnitude of biological responses to ocean biogeochemistry change, existing knowledge suggests that ocean biogeochemical changes could exert a major selective pressure upon species and have the capability to reorganize patterns of body size, abundance, distribution, species richness, and ecosystem functioning (Table 1). Biological and ecological responses are likely to be magnified, especially if in interaction with other stressors [10,28], as there will be a need for multiple physiological adaptations. The expected biological response is further highlighted by the biological changes already observed in certain monitored ecosystems in response to recent environmental change. Coral reefs, in which massive bleaching and growth reduction have been linked to relatively minor contemporary warming and acidification [3,57], provide an excellent example of this. Even deep-sea ecosystems, for which the magnitude of biogeochemical shifts will be smaller (dotted lines in Figure 2E-G), may undergo substantial biological responses, mainly because the deep ocean is much more stable, and thus its faunas are likely adapted to narrower ranges of environmental variation than those in shallow marine habitats [14,58]. We reemphasize that a standing challenge is to determine

the preponderance of taxa from different marine habitats and ecosystems that will be sensitive to ocean biogeochemistry change.

Vulnerability of Coastal People to Ocean Biogeochemistry Change

Here we quantified the relative vulnerability of coastal people to ocean biogeochemistry change in the traditional sense of exposure to environmental change, dependency of potentially impacted ocean goods and services, and social adaptability [34,35,44,45].

We determined the level of exposure of each Exclusive Economic Zone in the world to the cumulative negative ocean biogeochemistry change, on a scale ranging between 0 (i.e., no ocean biogeochemistry change) and 4 (i.e., maximum observed biogeochemistry change in all four analyzed parameters) (data from Figure 3B; we analyzed only negative changes given their overwhelming coverage globally, and because those changes are likely to have the largest impacts on the supply of ocean goods and services). For the purpose of classification, cumulative negative biogeochemical changes were divided into three equal bins to classify countries with low, medium, and high exposure to ocean biogeochemistry change.

To quantify levels of dependency, we used three different metrics of peoples' dependence on the ocean: jobs, revenues, and food. Job dependency was measured as the fraction of the countries' work force employed by marine fishing, the marine tourism industry, mariculture, and marine mammal watching. Revenue dependency was measured as the fraction of a country's Gross Domestic Product (GDP) generated by revenues from marine tourism, fishing, mariculture, and marine mammal watching. Food dependency was the fraction of animal protein consumption supplied by seafood. All three dependencies were added and divided in three equal bins to indicate countries of low, medium, and high dependency.

Societal adaptability to environmental change was quantified as per capita GDP, assuming that richer countries will have more alternatives, higher capacity, and adaptability. For the purpose of classification, we defined low-, medium-, and high-income countries depending on whether annual per capita GDP was smaller than US\$4,000, between US\$4,000 and US\$12,000 and larger than US\$12,000, respectively (sources of data are presented in Table S6).

For each country, we estimated the number of coastal people (i.e., living within 50 km of the coast) within each category of exposure, dependency, and adaptability (we provide global summaries in the main text and detailed country results in the Supporting Information section). We found that approximately 1.4 billion people live in the coastal areas of countries whose Exclusive Economic Zones will experience medium to high ocean biogeochemistry change by 2100 under the RCP45. Of those, ~690 million live in countries with a medium to high ocean dependence, and of these \sim 470 million live in low-income countries (Figure 6A). The situation will be more dramatic under the RCP85, according to which 2.02 billion coastal people will live in countries with medium to high ocean biogeochemistry change; of those, 1.12 billion live in countries of medium to high ocean dependence; and of these, ~ 870 million live in low-income countries (Figure 6B; detailed statistics of the change in each biochemical parameter at each Exclusive Economic Zone and sources of error owing to accuracy and precision in the Earth System Models are shown in Table S5). These results highlight the considerable challenges for human adaptability likely to emerge from ocean biogeochemistry change. Not only does a considerable fraction of the world's human population constantly use resources that will be impacted by ocean climate change, but such people are also located in developing countries with low capacity for adaptation to climate change. This limited socioeconomic capacity could also hamper the ability to benefit from "positive" ecosystem changes, if such new opportunities require costly adaptability [38].

Concluding Remarks

Although a mechanistic model of how ocean biogeochemical changes alter biological and social systems will be difficult to develop, existing knowledge suggests that the responses to the exposure of expected ocean biogeochemical change could be considerable. First, the array of interrelated parameters affected by increasing CO₂ emissions provide a much more worrisome picture than consideration of single stressors alone, as most of the world's oceans will be influenced by changes in multiple biogeochemical parameters, and thus adaptation will require multiple physiological adjustments from marine species. Additionally, there is the potential for synergistic responses to co-occurring stressors, and indirect ecological releases and trophic cascades. Secondly, human dependence on marine goods and services is also substantial in countries that will experience considerable ocean biogeochemistry change, particularly among low-income countries. This highlights the looming vulnerability to climate change in developing/low-income countries, and an unfortunate disparity between those who benefit economically from the processes creating climate change and those who will have to pay most of the environmental and social costs. The kind of biogeochemical stressors identified here will be further compounded by sea level rise, which has already been identified as a major potential socioeconomic consequence from climate change. These results provide a refined and synoptic numerical projection of change in key biogeochemical parameters upon marine biota and human societies, and indicate that if global CO₂ emissions are not reduced, substantial degradation of marine ecosystems and associated human hardships are very likely to occur.

Methods

Our analysis builds on recent ocean physical and biochemical projections developed as part of the Coupled Model Intercomparison Project Phase 5 to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [47]. As of July 2012, there were 31 Earth System Models from 18 centers in nine countries that modeled at least one of the ocean parameters analyzed here (Table S1). For analysis, all parameters were interpolated into a common 1° by 1° grid (assessment of multiple interpolation methods is provided in the Supplement S1). In total, over 27,000 years of data from the different models and variables were processed. Given the number and size of the files, we used several tools to optimize data processing, which are made available in Supplement S2. To quantify the robustness of Earth System Models, we compared projections among models (to measure model precision) and with actual data (to measure model accuracy) (data sources are indicated in Table S6). The multimodel average projections in the different biogeochemical parameters, in response to the analyzed CO₂ scenarios, were overlapped with the distribution of different marine habitats and biodiversity hotspots to calculate how much individual and combined change will occur upon each habitat and hotspot (additional details are provide in Figure 5 and Table S4). Finally, for each Exclusive Economic Zone in the world, we calculated the projected cumulative change in all biogeochemical parameters analyzed here (Figure 3B), and quantified human vulnerability to this change by using country-level data on current social resilience (in terms of wealth and assuming that richer countries will have more alternatives, higher capacity, and adaptability) and human dependence for ocean goods and services arising from food, jobs,

and revenue (Results for individual countries are shown in Table S5 and data sources in Table S6.)

Supporting Information

Figure S1 Representative concentration pathways. (DOCX)

Figure S2 Absolute and scaled changes in temperature, oxygen, pH, and primary food availability to the year 2100 at the ocean surface and floor. (DOCX)

Table S1Models used and their individual results.(DOCX)

 Table S2
 Accuracy and precision of multi-model predictions.

(DOCX)

Table \$3Ocean biogeochemistry change by availablemodels.

(DOCX)

 Table S4
 Expected climate change on marine habitats and biodiversity hotspots.

 (DOCY)
 (DOCY)

(DOCX)

Table S5Climate change by Exclusive Economic Zoneand vulnerability of coastal people given their depen-dency of the ocean and wealth.(DOCX)

References

- Solomon S, Qin D, Manning M (2007) The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Geneva. 104 p.
- Hoegh-Guldberg O, Bruno JF (2010) The impact of climate change on the world's marine ecosystems. Science 328: 1523–1528.
- Hoegh-Guldberg O, Mumby PJ, Hooten AJ, Steneck RS, Greenfield P, et al. (2007) Coral reefs under rapid climate change and ocean acidification. Science 318: 1737–1742.
- Mann ME, Bradley RS, Hughes MK (1998) Global-scale temperature patterns and climate forcing over the past six centuries. Nature 392: 779–787.
- Wigley TML, Raper SCB (2001) Interpretation of high projections for globalmean warming. Science 293: 451–454.
- Zeebe RE, Zachos JC, Caldeira K, Tyrrell T (2008) Carbon emissions and acidification. Science 321: 51–52.
- Cao L, Caldeira K (2008) Atmospheric CO2 stabilization and ocean acidification. Geophysical Research Letters 35, L19609.
- Quéré LC, Andres RJ, Boden T, Conway T, Houghton RA, et al. (2012) The global carbon budget 1959–2011. Earth Syst Sci Data Discuss 5: 1107– 1157.
- Cocco V, Joos F, Steinacher M, Frölicher TL, Bopp L, et al. (2012) Oxygen and indicators of stress for marine life in multi-model global warming projections. Biogeosciences Discuss 9: 10785–10845.
- Doney SC, Ruckelshaus M, Duffy JE, Barry JP, Chan F, et al. (2012) Climate change impacts on marine ecosystems. Annu Rev Mar Sci 4: 11– 37.
- Steinacher M, Joos F, Frölicher TL, Bopp L, Cadule P, et al. (2010) Projected 21st century decrease in marine productivity: a multi-model analysis. Biogeosciences 7: 979–1005.
- 12. Royal Society of London (2005) Ocean acidification due to increasing atmospheric carbon dioxide. Royal Society of London.
- Hönisch B, Ridgwell A, Schmidt DN, Thomas E, Gibbs SJ, et al. (2012) The Geological Record of Ocean Acidification. Science 335: 1058–1063.
- Danovaro R, Dell'Anno A, Pusceddu A (2004) Biodiversity response to climate change in a warm deep sea. Ecology Letters 7: 821–828.
- Orr JC, Fabry VJ, Aumont O, Bopp L, Doney SC, et al. (2005) Anthropogenic ocean acidification over the twenty-first century and its impact on calcifying organisms. Nature 437: 681–686.
- Yasuhara M, Hunt G, Breitburg D, Tsujimoto A, Katsuki K (2012) Humaninduced marine ecological degradation: micropaleontological perspectives. Ecology and Evolution: 3242–3268.
- Peck LS, Clark MS, Morley SA, Massey A, Rossetti H (2009) Animal temperature limits and ecological relevance: effects of size, activity and rates of change. Functional Ecology 23: 248–256.

Table S6Source of data.(DOCX)

Supplement S1 Standardization of grid resolutions among models.

Supplement S2 R scripts for processing CMIP5 files.

(DOCX)

Acknowledgments

We thank multiple data providers for making their data freely available; their names are provided in Table S6. We acknowledge the World Climate Research Programme's Working Group on Coupled Modelling, which is responsible for CMIP, and we thank the climate modeling groups (listed in Table S1) for producing and making available their model output. For CMIP the United States Department of Energy's Program for Climate Model Diagnosis and Intercomparison provides coordinating support and led development of software infrastructure in partnership with the Global Organization for Earth System Science Portals.

Author Contributions

The author(s) have made the following declarations about their contributions: Conceived and designed the experiments: CM CLW AR TA ARB DB LB QC MC RD AJG BMG PRH JI DOBJ LAL HN KN ERL MR HAR CRS AKS ART JFT PU LW TW MY. Performed the experiments: CM AR. Analyzed the data: CM QC CLW. Wrote the paper: CM CLW AR TA ARB DB LB QC MC RD AJG BMG PRH JI DOBJ LAL HN KN ERL MR HAR CRS AKS ART JFT PU LW TW MY.

- Danovaro R, Dell'Anno A, Fabiano M, Pusceddu A, Tselepides A (2001) Deep-Sea ecosystem response to climate changes: the Eastern Mediterranean case study. Trends in Ecology & Evolution 16: 505–510.
- Ruhl H, Ellena J, Smith K (2008) Connections between climate, food limitation, and carbon cycling in abyssal sediment communities. Proceedings of the National Academy of Sciences 105: 17006–17011.
- Diaz RJ, Rosenberg R (2008) Spreading dead zones and consequences for marine ecosystems. Science 321: 926–929.
- Helly JJ, Levin LA (2004) Global distribution of naturally occurring marine hypoxia on continental margins. Deep Sea Research Part I: Oceanographic Research Papers 51: 1159–1168.
- Levin LA, Etter RJ, Rex MA, Gooday AJ, Smith CR, et al. (2001) Environmental influences on regional deep-sea species diversity. Annual Review of Ecology and Systematics 132: 51–93.
- 23. Gooday A, Levin L, Aranda da Silva A, Bett B, Cowie G, et al. (2009) Faunal responses to oxygen gradients on the Pakistan margin: A comparison of foraminiferans, macrofauna and megafauna. Deep Sea Research Part II: Topical Studies in Oceanography 56: 488–502.
- Stramma L, Schmidtko S, Levin LA, Johnson GC (2010) Ocean oxygen minima expansions and their biological impacts. Deep Sea Research Part I: Oceanographic Research Papers 57: 587–595.
- Iglesias-Rodriguez MD, Halloran PR, Rickaby REM, Hall IR, Colmenero-Hidalgo E, et al. (2008) Phytoplankton calcification in a high-CO2 world. Science 320: 336–340.
- Anthony KRN, Kline DI, Diaz-Pulido G, Dove S, Hocgh-Guldberg O (2008) Ocean acidification causes bleaching and productivity loss in coral reef builders. Proceedings of the National Academy of Sciences 105: 17442–17446.
- Pörtner H, Farrell A (2008) Physiology and climate change. Science 322: 690– 692.
- Caitlin C, Kristy K, Benjamin SH (2008) Interactive and cumulative effects of multiple human stressors in marine systems. Ecology Letters 11: 1304–1315.
- Mora C, Metzger R, Rollo A, Myers RA (2007) Experimental simulations about the effects of overexploitation and habitat fragmentation on populations facing environmental warming. Proceedings of the Royal Society B-Biological Sciences 274: 1023–1028.
- Darling ES, Côté IM (2008) Quantifying the evidence for ecological synergies. Ecology Letters 11: 1278–1286.
- Cheung WWL, Sarmiento JL, Dunne J, Frolicher TL, Lam VWY, et al. (2012) Shrinking of fishes exacerbates impacts of global ocean changes on marine ecosystems. Nat Clim Chang 3: 254–258.
- Wood HL, Spicer JI, Widdicombe S (2008) Ocean acidification may increase calcification rates, but at a cost. Proceedings of the Royal Society B: Biological Sciences 275: 1767–1773.

- Byrne M (2011) Impact of ocean warming and ocean acidification on marine invertebrate life history stages: vulnerabilities and potential for persistence in a changing ocean. Oceanogr Mar Biol Annu Rev 49: 1–42.
- Allison EH, Perry AL, Badjeck MC, Adger WN, Brown K, et al. (2009) Vulnerability of national economies to the impacts of climate change on fisheries. Fish and Fisheries 10: 173–196.
- Cinner J, McClanahan TR, Graham NAJ, Daw TM, Maina J, et al. (2012) Vulnerability of coastal communities to key impacts of climate change on coral reef fisheries. Global Environmental Change 22: 12–20.
- Merino G, Barange M, Blanchard JL, Harle J, Holmes R, et al. (2012) Can marine fisheries and aquaculture meet fish demand from a growing human population in a changing climate? Global Environmental Change 22: 795–806.
- 37. Burke L (2012) Reefs at risk revisited in the Coral Triangle: World Resources Institute.
- Marshall NA, Marshall PA, Tamelander J, Obura D, Malleret-King D, et al. (2010) A Framework for Social Adaptation to Climate Change: Sustaining Tropical Coastal Communities [sic] and Industries: IUCN.
- Costanza R, d'Arge RC, de Groot R, Farber S, Grasso M, et al. (1997) The value of the world's ecosystem services and natural capital. Nature 387: 253– 261.
- Whitehead H, McGill B, Worm B (2008) Diversity of deep-water cetaceans in relation to temperature: implications for ocean warming. Ecology Letters 11: 1198–1207.
- Cheung WWL, Lam VWY, Sarmiento JL, Kearney K, Watson R, et al. (2009) Projecting global marine biodiversity impacts under climate change scenarios. Fish and Fisheries 10: 235–251.
- 42. Cheung WW, Dunne J, Sarmiento JL, Pauly D (2011) Integrating ecophysiology and plankton dynamics into projected maximum fisheries catch potential under climate change in the Northeast Atlantic. ICES Journal of Marine Science: Journal du Conseil 68: 1008–1018.
- Sumaila UR, Cheung WWL, Lam VWY, Pauly D, Herrick S (2011) Climate change impacts on the biophysics and economics of world fisheries. Nature Climate Change 1: 449–456.
- Adger WN, Hughes TP, Folke C, Carpenter SR, Rockstrom J (2005) Socialecological resilience to coastal disasters. Science 309: 1036–1039.
- Smit B, Wandel J (2006) Adaptation, adaptive capacity and vulnerability. Global Environmental Change 16: 282–292.
- Caldeira K, E.Wickett M (2003) Anthropogenic carbon and ocean pH. Nature 425: 365.
- Taylor KE, Stouffer RJ, Meehl GA (2011) An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society 93: 485– 498.
- Vuuren DP, Edmonds JA, Kainuma M, Riahi K, Thomson AM, et al. (2011) The representative concentration pathways: an overview. Clim Chang 109: 5– 31.
- Stock CA, Alexander MA, Bond NA, Brander KM, Cheung WWL, et al. (2011) On the use of IPCC-class models to assess the impact of climate on living marine resources. Prog Oceanogr 88: 1–27.
- Reichler T, Kim J (2008) How well do coupled models simulate today's climate. Bulletin of the American Meteorological Society 89: 303–311.
- Le Treut H, Somerville R, Cubasch U, Ding Y, Mauritzen C, et al. (2007) Historical overview of climate change. Cambridge University Press. 95–127 p.
- Pierce DW, Barnett TP, Santer BD, Gleckler PJ (2009) Selecting global climate models for regional climate change studies. Proceedings of the National Academy of Sciences 106: 8441–8446.
- Meinshausen M, Smith SJ, Calvin K, Daniel JS, Kainuma MLT, et al. (2011) The RCP greenhouse gas concentrations and their extensions from 1765 to 2300. Climatic Change 109: 213–241.
- Tittensor DP, Mora C, Jetz W, Lotze HK, Ricard D, et al. (2010) Global patterns and predictors of marine biodiversity across taxa. Nature 466: 1098– 1101.
- Bellwood DR, Hughes TP, Folke C, Nystrom M (2004) Confronting the coral reef crisis. Nature 429: 827–833.
- Mora C, Aburto-Oropeza O, Ayala Bocos A, Ayotte PM, Banks S, et al. (2011) Global human footprint on the linkage between biodiversity and ecosystem functioning in reef fishes. PLoS Biol 9: e1000606.
- Hughes TP, Baird AH, Bellwood DR, Card M, Connolly SR, et al. (2003) Climate change, human impacts, and the resilience of coral reefs. Science 301: 929–933.
- Yasuhara M, Cronin T, deMenocal P, Okahashi H, Linsley B (2008) Abrupt climate change and collapse of deep-sea ecosystems. Proceedings of the National Academy of Sciences 105: 1556–1560.
- Halpern BS, Walbridge S, Selkoe KA, Kappel CV, Micheli F, et al. (2008) A global map of human impact on marine ecosystems. Science 319: 948–952.
- Clarke A, Johnston NM (1999) Scaling of metabolic rate with body mass and temperature in teleost fish. J Anim Ecol 68: 893–905.
- Hunt G, Roy K (2006) Climate change, body size evolution, and Cope's Rule in deep-sea ostracodes. Proceedings of the National Academy of Sciences of the United States of America 103: 1347–1352.
- Sheridan JA, Bickford D (2011) Shrinking body size as an ecological response to climate change. Nat Clim Chang 1: 401–406.
- Drinkwater KF (2005) The response of Atlantic cod (Gadus morhua) to future climate change. ICES J Mar Sci 62: 1327–1337.

- Manno C, Morata N, Bellerby R (2012) Effect of ocean acidification and temperature increase on the planktonic foraminifer *Neogloboquadrina pachyderma* (sinistral). Polar Biology 35: 1311–1319.
- Kroeker KJ, Kordas RL, Crim RN, Singh GG (2010) Meta-analysis reveals negative yet variable effects of ocean acidification on marine organisms. Ecology Letters 13: 1419–1434.
- Riebesell U, Schulz KG, Bellerby R, Botros M, Fritsche P, et al. (2007) Enhanced biological carbon consumption in a high CO2 ocean. Nature 450: 545–548.
- Hall-Spencer JM, Rodolfo-Metalpa R, Martin S, Ransome E, Fine M, et al. (2008) Volcanic carbon dioxide vents show ecosystem effects of ocean acidification. Nature 454: 96–99.
- Michaelidis B, Ouzounis C, Paleras A, Pörtner HO (2005) Effects of longterm moderate hypercapnia on acid-base balance and growth rate in marine mussels *Mytilus galloprovincialis*. Marine Ecology Progress Series 293: 109– 118.
- Barton A, Hales B, Waldbusser GG, Langdon C, Feely RA (2012) The Pacific oyster, *Crassostrea gigas*, shows negative correlation to naturally elevated carbon dioxide levels: Implications for near-term ocean acidification effects. Limnology and Oceanography 57: 698–710.
- Pörtner H-O (2008) Ecosystem effects of ocean acidification in times of ocean warming: a physiologist's view. Mar Ecol Prog Ser 373: 203–217.
- Levin L (2003) Oxygen minimum zone benthos: adaptation and community response to hypoxia. Oceanography and Marine Biology, an Annual Review 41: 1–45.
- Daufresne M, Lengfellner K, Sommer U (2009) Global warming benefits the small in aquatic ecosystems. Proceedings of the National Academy of Sciences 106: 12788–12793.
- Portner HO, Knust R (2007) Climate Change Affects Marine Fishes Through the Oxygen Limitation of Thermal Tolerance. Science 315: 95– 97.
- Wijgerde T, Jurriaans S, Hoofd M, Verreth JA, Osinga R (2012) Oxygen and heterotrophy affect calcification of the scleractinian coral *Galaxea fascicularis*. PLoS ONE 7: e52702.
- Smith CR, De Leo FC, Bernardino AF, Sweetman AK, Arbizu PM (2008) Abyssal food limitation, ecosystem structure and climate change. Trends in Ecology & Evolution 23: 518–528.
- Thiel H (1975) The size structure of the deep-sea benthos. Internationale Revue der gesamten Hydrobiologie 60: 575–606.
- Kaariainen JI, Bett BJ (2006) Evidence for benthic body size miniaturization in the deep sea. Journal of the Marine Biological Association of the United Kingdom 86, pp. 1339–1345.
- Rex MA, Etter RJ, Morris JS, Crouse J, McClain CR, et al. (2006) Global bathymetric patterns of standing stock and body size in the deep-sea benthos. Marine Ecology Progress Series 317, pp. 1–8.
- Schmidt DN, Renaud S, Bollmann J, Schiebel R, Thierstein HR (2004) Size distribution of Holocene planktic foraminifer assemblages: biogeography, ecology and adaptation. Marine Micropaleontology 50: 319– 338.
- Wigham B, Tyler P, Billett D (2003) Reproductive biology of the abyssal holothurian Amperima rosea: an opportunistic response to variable flux of surface derived organic matter? Journal of the Marine Biological Association of the UK 83: 175–188.
- Mora C, Ospina AF (2002) Experimental effect of cold, La Nina temperatures on the survival of reef fishes from Gorgona Island (eastern Pacific Ocean). Marine Biology 141: 789–793.
- Mora C, Ospina AF (2001) Tolerance to high temperatures and potential impact of sea warming on reef fishes of Gorgona Island (tropical eastern Pacific). Marine Biology 139: 765–769.
- McClain CR, Allen AP, Tittensor DP, Rex MA (2012) Energetics of life on the deep seafloor. Proceedings of the National Academy of Sciences 109: 15366– 15371.
- Vaquer-Sunyer R, Duarte CM (2011) Temperature effects on oxygen thresholds for hypoxia in marine benthic organisms. Glob Chang Biol 17: 1788–1797.
- Kelmo F, Hallock P (2013) Responses of foraminiferal assemblages to ENSO climate patterns on bank reefs of northern Bahia, Brazil: A 17-year record. Ecological Indicators 30: 148–157.
- Koch M, Bowes G, Ross C, Zhang XH (2013) Climate change and ocean acidification effects on seagrasses and marine macroalgae. Glob Chang Biol 19: 103–132.
- Syamsuddin ML, Saitoh S-i, Hirawake T, Bachri S, Harto AB (2013) Effects of El Nino-Southern Oscillation events on catches of Bigeye Tuna (*Thunnus obesus*) in the eastern Indian Ocean off Java. Fishery Bulletin 111: 175–188.
- Mora C (2008) A clear human footprint in the coral reefs of the Caribbean. Proceedings of the Royal Society B-Biological Sciences 275: 767–773.
- Mora C (2009) Degradation of Caribbean coral reefs: focusing on proximal rather than ultimate drivers. Reply to Rogers. Proceedings of the Royal Society B-Biological Sciences 276: 199–200.
- 90. Cerrano C, Bavestrello G, Bianchi C, Cattaneo-Vietti R, Bava S, et al. (2000) A catastrophic mass-mortality episode of gorgonians and other organisms in the

Ligurian Sea (North-western Mediterranean), summer 1999. Ecology Letters 3: 284–293.

- Harvell CD, Mitchell CE, Ward JR, Altizer S, Dobson AP, et al. (2002) Climate Warming and Disease Risks for Terrestrial and Marine Biota. Science 296: 2158–2162.
- Bruno JF, Selig ER, Casey KS, Page CA, Willis BL, et al. (2007) Thermal stress and coral cover as drivers of coral disease outbreaks. Plos Biology 5: 1220– 1227.
- Aronson RB, Bruno JF, Precht WF, Glynn PW, Harvell CD, et al. (2003) Causes of coral reef degradation. Science 302: 1502–1502.
- Widdicombe S, Spicer JI, Kitidis V (2011) Effects of ocean acidification on sediment fauna. In: Gattuso JP, Hansson LJ, editors. Ocean Acidification: Oxford University Press. pp. 176–191.
- Dupont S, Havenhand J, Thorndyke M (2008) CO2-driven acidification radically affects larval survival and development in marine organisms. Comparative Biochemistry and Physiology a-Molecular & Integrative Physiology 150: s-170.
- Kurihara H, Shimode S, Shirayama Y (2004) Sub-lethal effects of elevated concentration of CO2 on planktonic copepods and sea urchins. Journal of Oceanography 60: 743–750.
- Gibson R, Atkinson R, Gordon J, Smith I, Hughes D (2011) Impact of ocean warming and ocean acidification on marine invertebrate life history stages: vulnerabilities and potential for persistence in a changing ocean. Oceanogr Mar Biol Annu Rev 49: 1–42.
- Ginger KW, Vera CB, Dineshram R, Dennis CK, Adela IJ, et al. (2013) Larval and post-larval stages of Pacific oyster (*Crassostrea gigas*) are resistant to elevated CO2. PLoS ONE 8: e64147.
- Short FT, Neckles HA (1999) The effects of global climate change on seagrasses. Aquat Bot 63: 169–196.
- Neira C, Sellanes J, Levin LA, Arntz WE (2001) Meiofaunal distributions on the Peru margin: relationship to oxygen and organic matter availability. Deep Sea Research Part I: Oceanographic Research Papers 48: 2453–2472.
- 101. Gooday AJ, Bernhard JM, Levin LA, Suhr SB (2000) Foraminifera in the Arabian Sea oxygen minimum zone and other oxygen-deficient settings: taxonomic composition, diversity, and relation to metazoan faunas. Deep Sea Research Part II: Topical Studies in Oceanography 47: 25–54.
- Wishner K, Levin L, Gowing M, Mullineaux L (1990) Involvement of the oxygen minimum in benthic zonation on a deep seamount. Nature 346: 57–59.
- 103. Kuroyanagi A, da Rocha RE, Bijma J, Spero HJ, Russell AD, et al. (2013) Effect of dissolved oxygen concentration on planktonic foraminifera through laboratory culture experiments and implications for oceanic anoxic events. Marine Micropaleontology, 101, pp. 28–32.
- De Leo F, Drazen JC, Vetter EW, Rowden AA, Smith CR (2012) The effects of submarine canyons and the oxygen minimum zone on deep-sea fish assemblages off Hawai'i. Deep-Sea Research I 64: 54–70.
- Purcell JE (2012) Jellyfish and ctenophore blooms coincide with human proliferations and environmental perturbations. Ann Rev Mar Sci 4: 209–235.
- Langlet D, Geslin E, Baal C, Metzger E, Lejzerowicz F, et al. (2013) Foraminiferal survival after long term experimentally induced anoxia. Biogeosciences Discussions 10: 9243–9284.
- Ekau W, Auel H, Poertner HO, Gilbert D (2010) Impacts of hypoxia on the structure and processes in pelagic communities (zooplankton, macro-invertebrates and fish). Biogeosciences 7: 1669–1699.
- Yasuhara M, Hunt G, Cronin TM, Hokanishi N, Kawahata H, et al. (2012) Climatic forcing of Quaternary deep-sea benthic communities in the North Pacific Ocean. Paleobiology 38: 162–179.
- 109. Gooday AJ (2003) Benthic foraminifera (Protista) as tools in deep-water palaeoceanography: environmental influences on faunal characteristics. Advances in Marine Biology 46: 1–90.
- Billett D, Bett B, Rice A, Thurston M, Galéron J, et al. (2001) Long-term change in the megabenthos of the Porcupine Abyssal Plain (NE Atlantic). Prog Oceanog 50: 325–348.
- 111. Billett D, Bett B, Reid W, Boorman B, Priede I (2010) Long-term change in the abyssal NE Atlantic: The 'Amperima Event' revisited. Deep Sea Research Part II: Topical Studies in Oceanography 57: 1406–1417.
- 112. Vetter EW, Smith CR, De Leo FC (2010) Hawaiian hotspots: enhanced megafaunal abundance and diversity in submarine canyons on the oceanic islands of Hawaii. Marine Ecology 31: 183–199.
- Kreveld S (1996) Northeast Atlantic late Quaternary planktic foraminifera as primary productivity and water mass indicators. Scripta Geologica 113: 23–91.
- 114. Tecchio S, Ramírez-Llodra E, Sardà F, Palomera I, Mechó A, et al. (2011) Drivers of deep Mediterranean megabenthos communities along longitudinal and bathymetric gradients. Mar Ecol Prog Ser 439:181–192
- 115. Wei C-L, Rowe GT, Escobar-Briones E, Nunnally C, Soliman Y, et al. (2012) Standing stocks and body size of deep-sea Macrofauna: Predicting the baseline of 2010 "Deepwater Horizon" oil spill in the northern gulf of Mexico. Deep Sea Research Part I: Oceanographic Research Papers.
- Perry AL, Low PJ, Ellis JR, Reynolds JD (2005) Climate Change and Distribution Shifts in Marine Fishes. Science 308: 1912–1915.
- 117. Yasuhara M, Hunt G, Cronin TM, Okahashi H (2009) Temporal latitudinalgradient dynamics and tropical instability of deep-sea species diversity. Proceedings of the National Academy of Sciences 106: 21717–21720.

- Comeaux RS, Allison MA, Bianchi TS (2012) Mangrove expansion in the Gulf of Mexico with climate change: Implications for wetland health and resistance to rising sea levels. Estuar Coast Shelf Sci 96: 81–95.
- Nesis KN (1997) Gonatid squids in the sunarctic North Pacific: ecology, biogeography, niche diversity and role in ecosystem. Adv Mar Biol 32: 243– 324.
- Narayanaswamy BE, Renaud PE, Duineveld GC, Berge J, Lavaleye MS, et al. (2010) Biodiversity trends along the western European margin. PLoS ONE 5: e14295.
- Hill JK, Griffiths HM, Thomas CD (2011) Climate change and evolutionary adaptations at species' range margins. Annual review of entomology 56: 143– 159.
- Mora C, Zapata FA (2013) Human footprints on biodiversity. In: Krohde K, editor. The Balance of Nature and Human Impact: Cambridge University. pp. 239–257.
- 123. Tittensor DP, Baco AR, Hall-Spencer JM, Orr JC, Rogers AD (2010) Seamounts as refugia from ocean acidification for cold-water stony corals. Marine Ecology 31: 212–225.
- Guinotte JM, Orr J, Cairns S, Freiwald A, Morgan L, et al. (2006) Will humaninduced changes in scawater chemistry alter the distribution of deep-sea scleractinian corals? Frontiers in Ecology and the Environment 4: 141–146.
- 125. Stramma L, Prince ED, Schmidtko S, Luo J, Hoolihan JP, et al. (2011) Expansion of oxygen minimum zones may reduce available habitat for tropical pelagic fishes. Nature Climate Change 2: 33–37.
- Prince ED, Goodyear CP (2006) Hypoxia-based habitat compression of tropical pelagic fishes. Fisheries Oceanography 15: 451–464.
- 127. Prince ED, Luo J, Phillip Goodycar C, Hoolihan JP, Snodgrass D, et al. (2010) Ocean scale hypoxia-based habitat compression of Atlantic istiophorid billfishes. Fisheries Oceanography 19: 448–462.
- Gilly WF, Beman JM, Litvin SY, Robison BH (2013) Oceanographic and biological effects of shoaling of the oxygen minimum zone. Annual Review of Marine Science 5: 393–420.
- 129. Koslow J, Goericke R, Lara-Lopez A, Watson W (2011) Impact of declining intermediate-water oxygen on deepwater fishes in the California Current. Marine Ecology Progress Series 436: 207–218.
- Schumacher S, Jorissen FJ, Dissard D, Larkin KE, Gooday AJ (2007) Live (Rose Bengal stained) and dead benthic foraminifera from the oxygen minimum zone of the Pakistan continental margin (Arabian Sea). Marine Micropaleontology 62: 45–73.
- Tittensor DP, Rex MA, Stuart CT, McClain CR, Smith CR (2011) Species– energy relationships in deep-sea molluscs. Biology Letters.
- Clarke A, Gaston KJ (2006) Climate, energy and diversity. Proceedings of the Royal Society B: Biological Sciences 273: 2257–2266.
- Currie DJ, Mittelbach GG, Cornell HV, Field R, Guegan JF, et al. (2004) Predictions and tests of climate-based hypotheses of broad-scale variation in taxonomic richness. Ecology Letters 7: 1121–1134.
- Allen AP, Brown JH, Gillooly JF (2002) Global biodiversity, biochemical kinetics, and the energetic-equivalence rule. Science 297: 1545–1548.
- Storch D (2003) Comment on "Global biodiversity, biochemical kinetics, and the energetic-equivalence rule". Science 299: 346.
- Cronin T, ME R (1997) Orbital forcing of deep-sea benthic species diversity. Nature 385: 624–627.
- Mora C, Robertson DR (2005) Causes of latitudinal gradients in species richness: A test with fishes of the Tropical Eastern Pacific. Ecology 86: 1771– 1782.
- Yasuhara M, Hunt G, van Dijken G, Arrigo KR, Cronin TM, et al. (2012) Patterns and controlling factors of species diversity in the Arctic Ocean. Journal of Biogeography 39: 2081–2088.
- Widdicombe S, Spicer JI (2008) Predicting the impact of ocean acidification on benthic biodiversity: what can animal physiology tell us? Journal of Experimental Marine Biology and Ecology 366: 187–197.
- 140. Widdicombe S, Dashfield S, McNeill C, Needham H, Beesley A, et al. (2009) Effects of CO2 induced seawater acidification on infaunal diversity and sediment nutrient fluxes. Mar Ecol Prog Ser 379: 59–75.
- Corliss B, Brown C, Sun X, Showers W (2009) Deep-sea benthic diversity linked to seasonality of pelagic productivity. Deep-Sea Research I 56: 835– 841.
- 142. Gooday AJ, Bett BJ, Shires R, Lambshead PJD (1998) Deep-sea benthic foraminiferal species diversity in the NE Atlantic and NW Arabian Sea: a synthesis. Deep-Sea Research Part II 45: 165–201.
- 143. Shi D, Xu Y, Hopkinson BM, Morel FMM (2010) Effect of ocean acidification on iron availability to marine phytoplankton. Science 327: 676– 679.
- Hofmann M, Schellnhuber H-J (2009) Oceanic acidification affects marine carbon pump and triggers extended marine oxygen holes. Proceedings of the National Academy of Sciences 106: 3017–3022.
- 145. Woulds C, Cowie GL, Levin LA, Andersson JH, Middelburg JJ, et al. (2007) Oxygen as a control on sea floor biological communities and their roles in sedimentary carbon cycling. Limnology and Oceanography 52, pp. 1698–1709.
- 146. Levin L, McGregor A, Mendoza G, Woulds C, Cross P, et al. (2013) Macrofaunal colonization across the Indian Margin oxygen minimum zone. Biogeosciences Discussions 10: 9451–9492.

- 147. van Oevelen D, Soetaert K, García R, de Stigter HC, Cunha MR, et al. (2011) Canyon conditions impact carbon flows in food webs of three sections of the Nazaré canyon. Deep Sea Research Part II: Topical Studies in Oceanography 58: 2461–2476.
- 148. Amaro T, Bianchelli S, Billett D, Cunha M, Pusceddu A, et al. (2010) The trophic biology of the holothurian *Molpadia musculus*: implications for organic matter cycling and ecosystem functioning in a deep submarine canyon. Biogeosciences 7: 2419–2432.
- 149. van Nugteren P, Herman PM, Moodley L, Middelburg JJ, Vos M, et al. (2009) Spatial distribution of detrital resources determines the outcome of competition between bacteria and a facultative detritivorous worm. Limnology and Oceanography 54: 1413.
- van Nugteren P, Moodley L, Brummer G-J, Heip CH, Herman PM, et al. (2009) Seafloor ecosystem functioning: the importance of organic matter priming. Marine Biology 156: 2277–2287.